Author’s response to reviews

Title: Human rabies post exposure prophylaxis at the Pasteur Institute of Dakar, Senegal: trends and risk factors

Authors:

Mamadou Korka Diallo (kdiallo@pasteur.sn)
Alpha Oumar Diallo (alphaomar2002@hotmail.com)
Anta Dicko (nadicko@pasteur.sn)
Vincent Richard (virc@hotmail.fr)
Emmanuelle Espié (emmanuelle.espie@gmail.com)

Version: 6 Date: 18 Mar 2019

Author’s response to reviews:

Dear Oliver Cocks,

Thank you for the consideration you have given to our manuscript “Human rabies post exposure prophylaxis at the Pasteur Institute of Dakar, Senegal: trends and risk factors”.

Please find below our detailed responses to your last comments.

We look forward to future correspondence regarding our resubmission and hope that this final version will suit you for an publication approval.

Sincerely yours,

Emmanuelle Espié, on behalf of all the co-authors
Editor Comments:

1. The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section.

   The authors prefer not to share all raw data, therefore we state the following justification “The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request” in the statement section “Data Availability”

2. Additional files not published alongside our manuscript, including cover/response letters have been removed

3. A single final clean version of our manuscript has been uploaded, as well as the two relevant figures in separate files.